Author’s response to reviews

Title: teratoma of lumbosacral region: a site of extreme rarity

Authors:

mohd faheem Dr (faheemjnmc@gmail.com)
HASAN H SYED DR (hasanharris@rediffmail.com)
DINESH KARDAM DR (dinesh.kardam@rediffmail.com)
VEENA MAHESHWARI DR (vegovil@rediffmail.com)
ROOBINA KHAN DR (roobinakhan@rediffmail.com)

Version: 2 Date: 8 February 2011

Author’s response to reviews:

To,
The Editor,
Journal of medical case report,
Sub: submission of revised manuscript
Sir,
It is to submit my revised manuscript entitled “TERATOMA OF LUMBOSACRAL REGION: A SITE OF EXTREME RARITY” (Manuscript ID 1860942143452148) with corrections as suggested by reviewers.
As pointed out by reviewer 1, the whole article has been grammatically corrected and highlighted with red ink where as a paragraph on differential diagnosis has been added as demanded by reviewer 2.
Thanking you
Yours
Dr. Mohd. Faheem
Corresponding author,
Deptt. Of surgery,
J.N.Medical College, Aligarh Muslim University,
Aligarh, India.
Email: faheemjnmc@gmail.com
Phone no: 9897783818